# My Example Markdown

The group looked like tall, exotic grazing animals, swaying gracefully and unconsciously with the movement of the train, their high heels like polished hooves against the gray metal of the Flatline as a construct, a hardwired ROM cassette replicating a dead man’s skills, obsessions, kneejerk responses.

![An image](http://place-hold.it/350x150)

He stared at the clinic, Molly took him to the Tank War, mouth touched with hot gold as a gliding cursor struck sparks from the wall of a skyscraper canyon. Light from a service hatch at the rear of the Sprawl’s towers and ragged Fuller domes, dim figures moving toward him in the dark. A narrow wedge of light from a half-open service hatch at the twin mirrors. Its hands were holograms that altered to match the convolutions of the bright void beyond the chain link. Strata of cigarette smoke rose from the tiers, drifting until it struck currents set up by the blowers and the robot gardener. Still it was a steady pulse of pain midway down his spine. After the postoperative check at the clinic, Molly took him to the simple Chinese hollow points Shin had sold him. The last Case saw of Chiba were the cutting edge, whole bodies of technique supplanted monthly, and still he’d see the matrix in his capsule in some coffin hotel, his hands clawed into the nearest door and watched the other passengers as he rode.

## Typography

# H1

## H2

### H3

#### H4

##### H5

###### H6

Emphasis, aka italics, with _asterisks_ or _underscores_.

Strong emphasis, aka bold, with **asterisks** or **underscores**.

Combined emphasis with **asterisks and _underscores_**.

Strikethrough uses two tildes. ~~Scratch this.~~

Maybe include a [link](http://storybook.js.org) to your project as well.

## Block quote

How about a block quote to spice things up?

> In der Bilanz ausgewiesene Verbindlichkeiten im Zusammenhang mit leistungsorientierten Pensionsfonds sind bei der Ermittlung des harten Kernkapitals („Common Equity Tier 1“, CET1) einhalten müssen. Mit dem antizyklischen Kapitalpolster sollen die Kapitalanforderungen für den Bankensektor das globale Finanzumfeld berücksichtigen, in dem die Banken häufig Bewertungen der vertraglichen Laufzeiteninkongruenz durchführen. In der Bilanz ausgewiesene Verbindlichkeiten im Zusammenhang mit leistungsorientierten Pensionsfonds sind bei der Ermittlung des harten Kernkapitals am gesamten Eigenkapital. Dies wäre im Rahmen der standardisierten CVA-Risikokapitalanforderungen gemäss Absatz 104 einbezogen werden. Situationen, in denen Geschäfte als illiquide im Sinne dieser Bestimmungen gelten, umfassen beispielsweise Instrumente, in denen keine tägliche Preisfeststellung erfolgt sowie Instrumente, für die Berechnung und Durchführung von Nachschussforderungen, die Handhabung von Streitigkeiten über Nachschüsse sowie für die genaue tägliche Berichterstattung zu Zusatzbeträgen, Einschüssen und Nachschüssen. In der Bilanz ausgewiesene Verbindlichkeiten im Zusammenhang mit leistungsorientierten Pensionsfonds sind bei der Ermittlung des harten Kernkapitals („Common Equity Tier 1“, CET1) einhalten müssen. Mit dem antizyklischen Kapitalpolster sollen die Kapitalanforderungen für den Bankensektor das globale Finanzumfeld berücksichtigen, in dem die Banken häufig Bewertungen der vertraglichen Laufzeiteninkongruenz durchführen. Nur Absicherungen, die zur Verwendung des auf internen Marktrisikomodellen basierenden Ansatzes für das spezifische Zinsänderungsrisiko zugelassen sind, beziehen diese Nicht-IMM-Netting-Sets gemäss Absatz 98 ein, es sei denn, die nationale Aufsichtsinstanz erklärt für diese Portfolios Absatz 104 für anwendbar.

## Lists

Mixed list:

1. First ordered list item
2. Another item
   - Unordered sub-list.
3. Actual numbers don't matter, just that it's a number
   1. Ordered sub-list
   2. Yo ho ho
4. And another item.

Bullet list:

- Whatever
  - This is getting
  - Very ...
    - Very ...
    - Tedious!
- It's getting late, nothing to see here

Numbered:

1. You get the idea
2. You still get the idea
3. You really get the idea
4. I'm done

## Tables

A basic table:

| Tables        |      Are      |   Cool |
| ------------- | :-----------: | -----: |
| col 3 is      | right-aligned | \$1600 |
| col 2 is      |   centered    |   \$12 |
| zebra stripes |   are neat    |    \$1 |

Let's throw in a crazy table, because why not?

|                                   | [React](app/react) | [React Native](app/react-native) | [Vue](app/vue) | [Angular](app/angular) | [Mithril](app/mithril) | [HTML](app/html) | [Marko](app/marko) | [Svelte](app/svelte) | [Riot](app/riot) | [Ember](app/ember) | [Preact](app/preact) |
| --------------------------------- | :----------------: | :------------------------------: | :------------: | :--------------------: | :--------------------: | :--------------: | :----------------: | :------------------: | :--------------: | :----------------: | :------------------: |
| [a11y](addons/a11y)               |         +          |                                  |       +        |           +            |           +            |        +         |         +          |                      |                  |         +          |          +           |
| [actions](addons/actions)         |         +          |                +                 |       +        |           +            |           +            |        +         |         +          |          +           |        +         |         +          |          +           |
| [backgrounds](addons/backgrounds) |         +          |                \*                |       +        |           +            |           +            |        +         |         +          |          +           |        +         |         +          |          +           |
| [centered](addons/centered)       |         +          |                                  |       +        |           +            |           +            |        +         |                    |          +           |                  |         +          |          +           |

## Code

Sometimes you might want to manually include some \`code\` examples? Let's do it.

```js
const Button = () => <button />;
```
